Author’s response to reviews

Title: From positive screen to engagement in treatment: a preliminary study of the impact of a new model of care for prisoners with serious mental illness

Authors:

Krishna Pillai (Krishna.Pillai@waitematadhb.govt.nzKrisha)
Paul Rouse (p.rouse@auckland.ac.nz)
Brian McKenna (Brian.McKenna@mh.org.au)
Jeremy Skipworth (Jeremy.Skipworthi@waitematadhb.govt.nz)
James Cavney (James.Cavney@waitematadhb.govt.nz)
Rees Tapsell (Rees.Tapsell@waikatodhb.health.nz)
Alexander Simpson (Sandy.Simpson@camh.ca)
Dominic Madell (Dominic.Madell@middlemore.co.nz)

Version: 2 Date: 02 Dec 2015

Author’s response to reviews:

December 2nd, 2015.

Associate Editor (Forensic Psychiatry),
BMC Psychiatry.

Dear Professor Kennedy,

Re: Manuscript "From positive screen to engagement in treatment: a preliminary study of the impact of a new model of care for prisoners with serious mental illness" (BPSY-D-15-00196).
Attached is a revised version of the manuscript for your consideration. We have outlined in a grid below the three further areas you wished us to address. Thank you and the reviewers for your comments which we believe have led to a refined manuscript. Not the yellow indicating the changes following the first review have been retained in the text.

We look forward to the outcome of this re-consideration.

Yours sincerely,

Brian McKenna,
Professor of Mental Health Nursing,
Australian Catholic University.

Comment:
The abstract uploaded separately in the submission system does not match the one in the manuscript. Please ensure that the most recent version of the abstract is in the manuscript, and that it is identical to the abstract in the submission system. Please also note that the comment about the trial registration is not necessary.

Response:
Corrected

Comment:
Please ensure that all authors are explicitly mentioned in this section.

Response:
Completed.
Comment:

Please remove the figures from the manuscript text document, and upload them instead as separate Figure items. Please do not include the figure legends in the image files but collect them instead at the end of the manuscript text document.

Response:

Completed.